UC Berkeley UC Berkeley Previously Published Works

Title

Increasing transmission of dengue virus across ecologically diverse regions of Ecuador and associated risk factors.

Permalink https://escholarship.org/uc/item/0948n7mw

Journal PLoS Neglected Tropical Diseases, 18(1)

Authors

Katzelnick, Leah Quentin, Emmanuelle Colston, Savannah <u>et al.</u>

Publication Date

2024

DOI

10.1371/journal.pntd.0011408

Peer reviewed



Citation: Katzelnick LC, Quentin E, Colston S, Ha T-A, Andrade P, Eisenberg JNS, et al. (2024) Increasing transmission of dengue virus across ecologically diverse regions of Ecuador and associated risk factors. PLoS Negl Trop Dis 18(1): e0011408. https://doi.org/10.1371/journal. pntd.0011408

Editor: Jason L. Rasgon, The Pennsylvania State University, UNITED STATES

Received: May 24, 2023

Accepted: January 15, 2024

Published: January 31, 2024

Copyright: This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the <u>Creative</u> Commons CC0 public domain dedication.

Data Availability Statement: The data sources and programs used for analyses are detailed in the Methods section. The code and data for the Fol models and the vector data have been deposited on Zenodo (10.5281/zenodo.10048448) and are freely available.

Funding: This research was supported by the Intramural Research Program of the National Institute of Allergy and Infectious Diseases (to LCK), the National Institutes of Health Grant RESEARCH ARTICLE

Increasing transmission of dengue virus across ecologically diverse regions of Ecuador and associated risk factors

Leah C. Katzelnick^{1*}, Emmanuelle Quentin², Savannah Colston³, Thien-An Ha⁴, Paulina Andrade⁴, Joseph N. S. Eisenberg³, Patricio Ponce⁵, Josefina Coloma^{4*}, Varsovia Cevallos^{5*}

1 Viral Epidemiology and Immunity Unit, Laboratory of Infectious Diseases, National Institute of Allergy and Infectious Diseases, National Institutes of Health, Bethesda, Maryland, United States of America, 2 Centro de Investigación en Salud Pública y Epidemiología Clínica (CISPEC), Facultad de Ciencias de la Salud Eugenio Espejo, Universidad UTE, Quito, Ecuador, 3 Department of Epidemiology, School of Public Health, University of Michigan, Ann Arbor, Michigan, United States of America, 4 Division of Infectious Diseases and Vaccinology, School of Public Health, University of California, Berkeley, Berkeley, California, United States of America, 5 Centro de Investigación en Enfermedades Infeciosas y Vectoriales (CIREV), Instituto Nacional de Investigación en Salud Pública (INSPI), Quito, Ecuador

* Leah.katzelnick@nih.gov (LCK); colomaj@berkeley.edu (JC); vcevallos@inspi.gob.ec (VC)

Abstract

The distribution and intensity of viral diseases transmitted by Aedes aegypti mosquitoes, including dengue, have rapidly increased over the last century. Here, we study dengue virus (DENV) transmission across the ecologically and demographically distinct regions or Ecuador. We analyzed province-level age-stratified dengue incidence data from 2000-2019 using catalytic models to estimate the force of infection of DENV over eight decades. We found that provinces established endemic DENV transmission at different time periods. Coastal provinces with the largest and most connected cities had the earliest and highest increase in DENV transmission, starting around 1980 and continuing to the present. In contrast, remote and rural areas with reduced access, like the northern coast and the Amazon regions, experienced a rise in DENV transmission and endemicity only in the last 10 to 20 years. The newly introduced chikungunya and Zika viruses have age-specific distributions of hospital-seeking cases consistent with recent emergence across all provinces. To evaluate factors associated with geographic differences in DENV transmission potential, we modeled DENV vector risk using 11,693 Aedes aegypti presence points to the resolution of 1 hectare. In total, 56% of the population of Ecuador, including in provinces identified as having increasing DENV transmission in our models, live in areas with high risk of Aedes aegypti, with population size, trash collection, elevation, and access to water as important determinants. Our investigation serves as a case study of the changes driving the expansion of DENV and other arboviruses globally and suggest that control efforts should be expanded to semi-urban and rural areas and to historically isolated regions to counteract increasing dengue outbreaks.

R01Al323721 (to JE and JC), and the Secretary of Higher Education, Science, Technology and Innovation (SENESCYT) Grant PIC-12-INH-002 (to PP and VC). LCK was supported in part by the Global Health Equity Scholars Program NIH Fogarty International Center training grant D43 TW010540. The funders had no role in the study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Author summary

The factors driving the increasing burden of arboviruses like dengue virus are not fully understood. In this study, we measured changes in dengue virus transmission intensity and arbovirus disease risk across Ecuador, an ecologically and demographically diverse South American country. We found that differences in the distribution of dengue cases could be explained by changes in transmission of dengue virus over time: transmission was limited to coastal provinces with large cities between 1980–2000, expanding thereafter to higher elevation areas and ecologically suitable but previously geographically and socially isolated provinces. We also used species distribution mapping to show that both urban and rural areas in Ecuador are at medium to high risk for *Aedes aegypti* presence, with population size, trash removal, elevation, and access to water as strong predictors. Our investigation reveals changes driving the expansion of dengue virus and other arboviruses globally and provides an approach for identifying areas at early stages of establishing endemic transmission that should be targeted for intense preventative efforts to avert future epidemics.

Introduction

Arboviruses transmitted by the *Aedes aegypti* mosquito (Culicidae), including the flaviviruses yellow fever virus, dengue viruses 1–4 (DENV1-4), and Zika virus (ZIKV) and alphavirus chikungunya virus (CHIKV), infect millions of people globally each year and cause a spectrum of life-threatening diseases with long-term sequelae, including hemorrhagic fevers, arthritis, and severe congenital abnormalities [1,2]. In the 19th and 20th centuries, explosive epidemics of arboviral disease primarily affected large population centers in tropical regions but in the 20th into the 21st century, arboviruses transmission worsened in previously affected areas and expanded into new regions [3–5]. As a result, dengue is one of the few tropical diseases with an increasing global burden over the last 20 years [6]. The distribution of other arboviruses has also increased over the last ten years as ZIKV and CHIKV were introduced to the Americas and caused back-to-back continent-wide pandemics [7,8].

Tracking factors driving the global expansion of arboviruses like DENV is limited by the quality of disease surveillance. In many regions, dengue incidence remains low but whether this is attributable to low reporting rates as opposed to low disease burden is difficult to determine. Recent and historical work suggests the force of infection (FoI), defined as the per-capita rate at which susceptible individuals become infected in a population within a given period, can be directly estimated from age-stratified incidence data and provides a useful measure of transmission intensity [9–14]. The age distribution of incidence is a more accurate measure of transmission intensity than aggregate case count data, and in general reflects transmission intensity estimates made using gold-standard seroprevalence data [9,12,15]. When longitudinal age-stratified seroprevalence or incidence data are available, it is possible to estimate changes in the FoI over time, as originally demonstrated for Hepatitis A [16] and numerous times for DENV [11,13,17–21].

The drivers of arboviral disease can be studied within Ecuador, an ecological and demographically diverse South American country with multiple distinct regions with reported arbovirus transmission [22]. Previous studies in urban centers in Ecuador have shown that household water access and storage methods, crowding, housing condition, garbage collection, and knowledge of dengue are strong local predictors of *Aedes aegypti* pupae presence and density as well as dengue cases [23–27]. Even within these urban areas, hyper-local hotspots have been identified, indicating not all neighborhoods are similarly affected each year [24,28,29]. Dengue cases and vector densities in both urban and rural communities along the Eastern coastal region of Ecuador are affected by seasonal and inter-annual climate effects, with strong predictors including precipitation and minimum temperature [23–25]. Fewer studies have been conducted in the more remote Amazon provinces of Ecuador, although vector mapping studies have suggested suitable conditions for *Aedes aegypti* in parts of this region [30].

Ecuador has a long history of arbovirus transmission, with the first case of yellow fever recorded in 1740 and the first outbreak in 1842 [31,32]. Between 1946 and 1970, eradication campaigns targeted the vectors for yellow fever and malaria, resulting in a major reduction in *Aedes* populations. *Aedes aegypti* soon returned, followed by the re-introduction of the four DENV serotypes [3]. National dengue outbreaks occurred in Ecuador in 1996, 2000, 2005, 2010, 2012, and 2014 and since 2000, the four DENV serotypes have been detected in all regions [33,34]. Phylogenetic analyses have shown both recent and historical introductions of DENV1, DENV2, and DENV4 in Ecuador that likely originated from Venezuela and Colombia, while chikungunya was likely introduced through the Caribbean [35–37].

Here, we modeled age-stratified dengue incidence data from 2000–2019 to estimate changes in the DENV transmission intensity for nearly 80 years and across provinces in Ecuador. We complement this analysis by evaluated factors associated with geographic differences in vector suitability in the last ten years by modeling 11,693 *Aedes aegypti* presence points. We identify provinces at different stages of establishing endemic DENV transmission as well as areas with high vector risk.

Materials and methods

Ethics statement

Human subjects data sets were from anonymized public health surveillance data collected by the Ecuadorian Ministry of Health as part of routine care. The Instituto Nacional de Estadística y Censos (INEC) data used in this study is freely available from an institutional website and is completely anonymous [38]. According to the international good clinical and research practice and in accordance with Ecuadorian legislation on clinical investigation, ethical approval by an institutional review board was not required. The public surveillance data collected by the Ecuador Ministry of Health (Sistema Nacional de Vigilancia en Salud Pública Ecuador, ViEpi data [39,40]) was made available for research purposes without confidential information (anonymous), under the "Organic Law of Transparency and Access to Public Information" (Article 4 of Ley Orgánica de Transparencia y Access a la Información Pública, LOTAIP, Law 24 of Article 81 of the Ecuadorian Constitution).

Geography, climate, and demographic characteristics of Ecuador

Ecuador is a country in South America of 283,561 square kilometers and a total population size of just over 17 million people. Ecuador is divided down the center by the Andes mountains; provinces in this region are at an altitude too high to support *Aedes* populations and DENV transmission (Fig 1A). To the west of the Andes are the coastal lowlands, which border the Pacific Ocean, are densely populated, have high temperatures and humidity, and distinct wet and dry seasons. The provinces of Esmeraldas, Manabí, Santa Elena/Guayas, Los Ríos and El Oro consist primarily of lowland regions while Santo Domingo de los Tsáchilas/Pichincha contain both lower and higher elevation areas. Esmeraldas is the only province north of the equator, and both Esmeraldas and higher-elevation Tsáchilas/Pichincha have lower and more stable year-round temperatures than other provinces in this region (Fig 1B). In all coastal provinces, the rainy season extends from November through May, with the highest number of



Fig 1. Geography and arbovirus seasonality of Ecuador. (A) Map of Ecuador, colored to indicate geographic and climate regions (coast, mountains, vs. Amazon). The base layer of the map is publicly provided by INEC [41]. (B) Temperature (grey dots and dotted lines), precipitation (dots colored by region, solid black line), and monthly distribution of arbovirus cases (black histograms). Annual province-level temperature and precipitation data from 2005–2019 were derived from the Universidad Tecnológica Equinoccial (UTE)/Centro de Investigación en Salud Pública y Epidemiología Clínica (CISPEC) database. Trend lines were fitted using local weighted regression (LOESS) models [42,43]. Monthly arbovirus case distributions for dengue, Zika, and chikungunya from 2014 to 2017 are derived from the ViEpi dataset.

arboviral disease cases occurring between March and July. Six of the eight cities in Ecuador with populations larger than 200,000 are on the coast, including the major port cities Guayaquil (Guayas) and Manta (Manabí). The six eastern provinces of Ecuador are part of the Amazon, with a tropical climate but lower population density and no cities with more than 120,000 people. Precipitation is high year-round, with slightly more rainfall from March to July and in the northern compared to the southern Amazon provinces. Consistently, there is less seasonality of arbovirus transmission in the Amazon than the coastal provinces. The Galapagos Islands constitute the fourth region of Ecuador and while cases are sporadically reported, there is little evidence of sustained transmission.

Age-stratified dengue incidence data and catalytic models

Dengue incidence was estimated from national surveillance data collected by INEC. The INEC dataset included all discharge data for hospitalized cases classified as vector-borne viral diseases and/or hemorrhagic fevers by International Statistical Classification of Diseases (ICD). Data were available from 2000 to 2019 and were grouped by age, sex, and province. For analysis, ICD codes A90 (Classic Dengue Fever) and A91 (Dengue Hemorrhagic Fever) were grouped as dengue cases (1997 WHO criteria for severe dengue disease). Data were binned into 18 age groups (<1, 1–4, 5–9, 10–14, 15–19, 20–24, 25–29, 30–34, 35–39, 40–44, 45–49, 50–54, 54–59, 60–64, 65–69, 70–74, 75–78, and \geq 80 years of age) for analysis. To estimate dengue incidence, we used provincial demographic data as our denominator. National census data from 1991, 2000, and 2010 were available for each province and age group; we used interpolation to estimate the age-specific population size for each year from 2000 to 2019. There were sufficient case counts in all provinces in the coastal region (Esmeraldas, Manabí, Guayas/Santa Elena, El Oro, Pichincha/Santo Domingo de los Tsáchilas, Los Ríos) to enable province-level

DENV FoI estimation. Although Santa Elena was separated from Guayas province and Santo Domingo de los Tsáchilas was separated from the Pichincha province in 2007, we leave them as combined provinces in our analyses to maintain consistency across years. Due to lower absolute case counts for the Amazon provinces (a median of < 4 cases per age group and year, leading to high noise in the data), we combined the case data for Sucumbíos, Napo, Orellana, Pastaza, Morona Santiago, and Zamora Chinchipe provinces. To evaluate the relative contribution of each age group to the total yearly incidence, we plotted the age-stratified dengue incidence normalized by the total incidence in each year to account for differences in absolute epidemic magnitude. Data were grouped into two periods (2000–2009 and 2010–2019) to visualize temporal changes in transmission. To compare distributions, data were fit with a LOESS curve with 95% confidence intervals. We used the age-adjusted dengue incidence data to estimate medians and interquartile range of ages by province for each 10-year period.

We used catalytic models to estimate historical and current FoI of DENV for each province based on age-stratified incidence using the method and code described by Rodriguez-Barraquer et al. [9], as follows. Given that cases were not serotyped but all four serotypes are known to circulate in Ecuador, we assumed equal FoI for the four DENV serotypes, as in previous dengue incidence models where case data by serotype were not available [9,12]. Thus, the fraction susceptible to all four DENV serotypes at age *a* and time *t* is:

$$x(a,t) = e^{\int_0^a -4\lambda(t-\tau)d}$$

Where $\lambda(t)$ is the average serotype-specific FoI and τ is the time since birth. We assumed the surveillance data reflect incidence of second DENV infections, as has been assumed previously using similar models [9,11,13,15]. This assumption is supported by decades of evidence that prior immunity to DENV is a strong risk factor for severe dengue disease in both Asia and the Americas [44–46], and that those experiencing second infections are much more likely to be hospital-seeking than primary or tertiary/quaternary infections and thus more likely to be detected by the surveillance system [18,47]. Models testing the relative contribution of primary, secondary, tertiary, and quaternary infections to case distributions further support this assumption [12,14]. We thus assume the at-risk population consists of individuals with primary DENV immunity, i.e., individuals infected with one DENV serotype but still at risk of infection with any of the other three serotypes. The fraction that has been infected with one serotype and remain susceptible to the other four serotypes is:

$$z_1(a,t) = 4\left(e^{-\int_0^a 3\lambda(t- au)d au}
ight)\left(1-e^{-\int_0^a \lambda(t- au)d au}
ight)$$

Thus, the age-specific incidence for DENV infection is assumed to reflect the primary immune group experiencing their second infection with any of the other serotypes:

$$I(a,t) = 3\lambda(t)z_1(a,t)$$

The expected number of reported cases of dengue at age *a* and time *t*, $\Lambda(a, t)$, is a function of age-specific infection incidence I(a,t), as well as the population size P(a,t) and reporting rate $\phi(t)$:

$$\Lambda(a,t) = I(a,t)P(a,t)\phi(t)$$

The reporting rate accounts for annual differences in the reporting rate of cases (not all cases are reported clinically) and the proportion of cases that are symptomatic among all DENV infections (not all infections cause cases). The variable $\phi(t)$ is assumed to be constant

across ages within a given year. We assume the observed number of reported cases at age a and time t, C(a,t), follows a Poisson distribution with the likelihood:

$$\mathcal{L}(C|\lambda,\phi) = \prod_t \prod_a rac{\Lambda(a,t)^{C(a,t)} e^{-\Lambda(a,t)}}{C(a,t)!}$$

Models were fit with Rstan [48] using the Bayesian Markov Chain Monte Carlo framework in R [49]. We ran four independent Markov chains with 10,000 iterations of warmup followed by 20,000 iterations for estimating parameters. Parameters were evaluated for convergence using the split \hat{R} statistic (values of 1 were assumed to indicate convergence) as well as visual inspection. All FoI estimates are shown as means with 95% credible intervals. The original authors of this code [9] note that given the large numbers for incidence, credible intervals may be systematically underestimated.

For our didactic example of CHIKV and ZIKV FoI, we assumed the age-specific incidence of chikungunya and Zika cases reflect the fully susceptible group experiencing a primary infection:

$$I(a,t) = \lambda(t)x(a,t)$$

All other equations are the same for CHIKV and ZIKV.

Comparison of dengue, chikungunya, and Zika incidence by province

The ViEpi dataset included de-identified individual case information for 2014 (November to December), 2015 (full year), 2016 (full year), and 2017 (January to April) meeting the following case definitions: (1) Dengue with Warning Signs or Severe Dengue, (2) chikungunya, and (3) Zika. The dataset included the following information for each case: age, sex, nationality, home district, province, canton, parish, date of illness, whether and how the case was confirmed (laboratory confirmation, clinical, contact case), hospitalization status, and outcome of case. Cases were grouped by the same age categories as indicated above for the INEC dataset. We use suspected, and not just laboratory-confirmed cases, as there is an age-specific bias in the cases that were confirmed. Notably, the incidence of chikungunya and Zika cases in infants and for Zika, reproductive age individuals, is higher than would be expected, possibly due to greater confirmation in these age groups. We used the ViEpi dataset (n = 41,777 total cases) to estimate: 1) province-level differences in age-stratified disease incidence of dengue, chikungunya, and Zika (2015–2016 data), and 2) province-level total monthly arbovirus case distributions (2014–2017).

Entomological data

Aedes aegypti presence was measured as part of the INSPI-SENESCYT Project "National Vigilance and Early Warning System for Vector Control of Dengue-Yellow Fever" (SATVEC PIC-12-INH-002). Systematic data collection procedures were used to ensure a comprehensive representation of mosquito populations across different geographic regions, allowing for a more robust and accurate analysis of their distribution and characteristics. *Aedes aegypti* mosquito samples were collected from 2013 to 2018 at 1785 collection sites in Ecuador and encompassing 23 provinces and 43 cantons (the second-level subdivision below provinces). Sampling sites ranged in elevation from 0 to 1650 meters above sea level. In total 946 records correspond to collections during the rainy season and 839 during the dry season. Geographic coordinates were obtained for all collections. Both immature and adult mosquito samples were collected from field locations and transported to the laboratory for rearing and species identification.

Species distribution mapping using maximum entropy models

We used the confirmed vector presence data in maximum entropy models to evaluate the species distributions for Aedes aegypti across Ecuador using the MaxEnt [50]. The maximum entropy model uses environmental and social layers to estimate the probability of presence of a species within each raster unit to predict the distribution of a species across the map [51,52]. MaxEnt was selected because it requires only presence data for the species being studied, has been widely used and produces consistent results even with a limited number of presence points, and has a GIS interface within the TerrSet GIS software (Clark Labs). This method has been used previously for characterizing the distribution of vectors like Aedes albopictus, Aedes aegypti, and arboviral diseases including dengue [53-56]. Variables previously identified as predictors of arbovirus transmission were included as layers in the model: two physiographic variables (elevation, slope), eight demographic and social variables derived from available census data (literacy, employment, overcrowding, population density, housing condition, water access, sewerage, garbage collection), and three environmental/climatic variables (precipitation, Daytime Land Surface Temperature, and the Enhanced Vegetation Index). All variables were downscaled at a resolution of 1 hectare (100m x 100m), corresponding to the most detailed layers in the INEC census database. In contrast with other models in Ecuador that have used the WorldClim database [30,55], we derived our environmental geodatabase directly from EarthData [57]. To process the spatiotemporal series of monthly images from 2010 to 2018 as environmental variables for analysis in the MaxEnt model, we applied an orthogonal transformation of n-dimensional image data using principal component analysis (PCA). This produced a new set of images (components) that are uncorrelated with one another and ordered with respect to the amount of variation (information) they represent from the original image set. These techniques were implemented in TerrSet. As we were working with a cube X-Y-T, the PCA could be adapted to the various dimensions. For the three climatic parameters (temperature, precipitation, and vegetation) we used two PCA components in T-mode and 2 PCA loadings in S-mode. Using this approach, the model was not affected by layers that would not be explicative.

We trained the MaxEnt model on 75% of presence points, with 25% for validation, in 10 bootstrap replicates each of 500 iterations. Model goodness-of-fit and prediction was evaluated using receiver operating characteristic (ROC) analyses and binomial tests of omission. The area under the ROC curve (AUC) was 0.92 for the training set and 0.916 for the test set, with a standard deviation of 0.003. For our binomial test of omission, we applied a logistic threshold of 0.416 that equals training sensitivity and specificity, yielding a fractional predicted area (used instead of the commission rate because there is no absence data) of 0.132, a training omission rate of 0.132, and a test omission rate is 0.156.

Results

Age-distributions of dengue incidence vary by province and over time

We measured temporal and geographic differences in DENV transmission intensity by estimating age-stratified dengue incidence for each province between 2000–2019. We estimated dengue incidence using national surveillance of Dengue Fever and Dengue Hemorrhagic Fever cases and interpolated age-specific province-level population size data. For data visualization, data were grouped into two 10-year periods, 2000–2009 and 2010–2019 (Fig 2). The median age of the population in each province ranged from 20–26 throughout the 20-year period, with the lowest median age in the Amazon and highest age in Guayas/Santa Elena and Pichincha/Santo Domingo (Table 1). The first decade of dengue incidence data had different



Fig 2. Age-specific incidence of dengue, normalized by year, for each province and period. Data are fitted using LOESS regression with 95% confidence intervals. Data are from the INEC database.

features compared to the second decade of data. For example, from 2000 to 2009, the population-dense coastal provinces of Guayas/Santa Elena, El Oro, and Manabí had the lowest median age of dengue cases (28–36 years) whereas in provinces with higher elevation or that were more rural and inaccessible (Esmeraldas and the Amazon), the median dengue incidence occurred at older ages (41–43 years). All provinces except Guayas/Santa Elena experienced a drop in median dengue incidence between 2000–2009 and 2010–2019, with the largest change in Manabí (by 17 years), as well as large changes in Esmeraldas, the Amazon, and Pichincha/ Santo Domingo (6–9 years).

Models of age-stratified dengue incidence reveal historical and geographic changes in DENV transmission

We assumed that the age distributions of dengue incidence by province were due to both current and historical transmission of DENV in each region. We estimated the FoI of DENV for

Table 1. Median and	interquartile range (IQR) of population-adjusted dengue incidence by province, from 2000-
2009 and 2010-2019.	Median and IQR of the population during the same period are shown for reference.

Province	Incidence:	Incidence: median (IQR)		Population: median (IQR)	
	2000-2009	2010-2019	2000-2009	2010-2019	
Manabí	36 (13-58)	19 (8-48)	24 (12-41)	25 (12-43)	
Pichincha/Santo Domingo	43 (25-60)	35 (14–59)	26 (13-42)	26 (13-42)	
Esmeraldas	41 (25–59)	32 (14–56)	22 (11-39)	22 (11-40)	
Amazon	43 (27–58)	37 (22–57)	20 (9–36)	21 (10-37)	
Guayas/Santa Elena	28 (11–51)	28 (13-58)	26 (13-42)	26 (12-43)	
El Oro	32 (16–51)	28 (13-57)	25 (12-42)	26 (13-44)	
Los Ríos	41 (22-60)	39 (18-62)	24 (11-41)	25 (12-42)	

https://doi.org/10.1371/journal.pntd.0011408.t001



10-year increments back to 1941, covering the earliest period when individuals in the dataset were alive. We modeled annual age-stratified dengue incidence directly using Bayesian models based on the method described by Rodriguez-Barraquer et al. [9]. For many pathogens, the age-distribution of disease reflects the average age at which individuals experience their first infection [58]. However, secondary DENV infections are more likely to cause symptoms and disease severe enough to elicit a hospital visit than either primary or tertiary/quaternary infections [12,14,18,47]. Thus, most cases captured by surveillance likely represent secondary DENV infections, both in Asian and the Americas [13]. Consistent with prior dengue incidence analyses, we thus assume that the age-specific incidence of dengue is driven by those experiencing their second DENV infection [9,11–15,44–47]. We observed that all provinces and regions had minimal DENV transmission for the period of 1941 to 1979 (Fig 3). Consistent with this observation, from 1946 to 1970, continent-wide Aedes eradication campaigns were underway. A program was implemented in 1958 to specifically monitor Aedes aegypti [3]. No dengue cases were recorded in Ecuador during this time, but Aedes successfully reinfested the cities in Guayas and Manabí between 1977 and 1985. Our FoI models estimated that between 1980 to 1989, DENV transmission intensity began to rise in Guayas/Santa Elena and El Oro (Fig 3). Then from 1990-1999, high DENV transmission was observed in Guayas/Santa Elena, El Oro, and Manabí. These results are consistent with the first documentation of dengue in Ecuador in 1985 and major outbreaks in Guayas in 1988, 1990, and 1993. At this time, dengue was re-emerging across the Americas, and the introduction of multiple serotypes resulted in the first observed cases of Dengue Hemorrhagic Fever/Dengue Shock Syndrome [3,4,59]. The largest and earliest dengue outbreaks were concentrated in the large coastal cities in Guavas and Manabí, consistent with our estimated timeline for the re-introduction of DENV to Ecuador.

Other regions with suitable ecological conditions for DENV transmission in Ecuador had not recorded major epidemics. Consistent with these epidemiological data, we found that the early rise of DENV in the southern coastal provinces was not observed in the northern coastal province of Esmeraldas, the higher elevation coastal provinces of Los Ríos and Pichincha/ Santo Domingo de los Tsáchilas, or the Amazon provinces, which had very low levels of DENV transmission from 1990–1999 (FoI: \leq 10 per 1000 people). These provinces had fewer dense urban settings and were more isolated before the year 2000, with limited road infrastructure. Pichincha/Santo Domingo, Los Ríos, and Esmeraldas had high DENV transmission intensity in 2000–2009 and 2010–2019 (FoI: 40–50 per 1000 people). The Amazon provinces were most delayed, experiencing a rise in DENV transmission only in the last 10 years (40 per 1000 people). Thus, we observe the provinces of Ecuador at different stages of endemicity, with some only recently experiencing increases in transmission intensity.

Age-distributions of recently emerged arboviral diseases reflect risk across ages

Chikungunya caused a major outbreak in Ecuador in 2015, soon followed by a national Zika epidemic in 2016 [8,35]. We expected that the age distributions of these new arboviral diseases would be similar across provinces, consistent with their recent introduction. We measured age and disease-specific incidence using data from a separate surveillance system, ViEpi, consisting of laboratory-confirmed and suspected cases of chikungunya, Zika, and Dengue with Warning Signs/Severe Dengue observed between 2015 and 2016. Across provinces, Zika and chikungunya incidence was mid-aged centered, consistent with recent emergence, although in some regions, Zika incidence was more concentrated in reproductive-age individuals, likely due to a bias in laboratory confirmation and reporting of cases (Fig 4). Notably, the age-distributions of chikungunya and Zika incidence more closely resembled the age distributions of dengue incidence in Esmeraldas, Los Ríos, Pichincha/Santo Domingo, and the Amazon, but were starkly different in Guayas/Santa Elena, El Oro, and Manabí. These findings further support the hypothesis that heterogeneity in the age distribution of dengue incidence across Ecuador reflects historical differences in the time of DENV emergence and the transition to endemicity.



Fig 4. Age-specific incidence of laboratory-confirmed and suspected Dengue with Warning Signs/Severe Dengue, chikungunya and Zika for each province in 2015–2016. Data are fitted using LOESS regression with 95% confidence intervals. In some provinces, unusual distributions for Zika suggest case-confirmation focused on specific age groups.

https://doi.org/10.1371/journal.pntd.0011408.g004



Fig 5. Age-stratified incidence of different arboviral diseases by time since emergence. Expected changes in age-specific incidence for a viral disease that emerged 20 years ago with an annual FoI of 50/1000, assuming the first infection causes disease (like chikungunya or Zika) vs. when there are four circulating viruses that emerged 20 years ago each with an annual FoI of 50/1000 and the second infection causes disease (like dengue).

The age-specific incidence distribution for arboviruses such as chikungunya and Zika, where a first infection causes disease, is expected to differ from dengue, where a second infection is most likely to cause disease, as illustrated using a theoretical modeling example (Fig 5). For all three diseases, as years pass since emergence, the age distribution of incidence shifts from mid-age distributed to younger ages as older individuals become immune and disease only affects young, naïve individuals. While for chikungunya and Zika, explosive epidemics are observed immediately across age groups, for dengue, it takes multiple years before high incidence is observed across age groups as individuals experience second infections. Further, the shape of the age distribution of cases at endemicity is distinct for dengue compared with chikungunya and Zika, reflecting these differences in the susceptible population.

Determinants of *Aedes* risk help explain geographic variation in DENV transmission intensity

To evaluate which environmental and demographic factors were associated with the observed geographic variation in DENV transmission intensity, we used maximum entropy models to identify predictors of *Aedes aegypti* presence data. We sampled immature and adult stages of *Aedes aegypti* mosquitoes in 1785 distinct collection sites for a total of n = 11,693 distinct samples (Fig 6). The percentage of sites with presence of *Aedes aegypti* in the coastal provinces was higher (35.9%) than in the Amazon provinces (13.4%) but *Aedes aegypti* was present at higher altitudes in the eastern Amazon basin (1650 meters) than the coastal region (1000 meters). In our maximum entropy models, 55.5% (123,087 km²) of the territory was suitable (medium or high risk) for *Aedes aegypti*. Most of the coastal lowlands had medium risk for *Aedes aegypti*, with high-risk zones in more populated areas. Almost all of Guayas and El Oro were at high



Fig 6. Maximum entropy geographic models using environmental and demographic variables to predict *Aedes aegypti* presence data. The base layer of the map is publicly provided by INEC [41].

risk, while large sections of Manabí, Esmeraldas, Pichincha/Santo Domingo, and Los Ríos were also at high risk. Much of the Amazon was predicted to be at low risk, although the northern Amazon and Amazon regions closer to the Andes were also at medium to high risk for *Aedes aegypti*. The variables that most strongly predicted *Aedes aegypti* presence included population density (63.3%), garbage collection (14.3%), and elevation (11.2%) (Table 2). *Aedes aegypti* risk was associated to a lesser extent with water access, vegetation, sewage connection, precipitation, house quality, and other variables.

Discussion

In this study, we described changes in the transmission intensity of DENV across ecologically diverse provinces of Ecuador over nearly 80 years and identified multiple environmental, demographic, and social factors associated with the distribution of the vector. Our estimates of the FoI of DENV and risk for the DENV vector complement and enrich the documented

Aedes model					
Variable	Percent contribution				
Population	63.3				
Garbage collection	14.3				
Elevation	11.2				
Water access	3.6				
Vegetation	1.8				
Sewage connection	1.6				
Precipitation	1.0				
Housing quality	0.7				
Other variables	2.5				

Table 2. Variable importance of risk factors predicting Aedes aegypti presence data.

https://doi.org/10.1371/journal.pntd.0011408.t002

record of DENV emergence across Ecuador and inform changes driving the expansion of arboviruses globally.

We found that DENV transmission in Ecuador increased as dengue resurged across the Americas, first in the southern coastal provinces with major port cities, then to other connected regions in the coastal lowlands. Consistent with our findings, a study in Brazil using catalytic FoI models documented a shift in dengue incidence from adults to children in the decades following emergence, a pattern expected of dengue as it becomes endemic [13]. Our analyses suggest that DENV only spread to more isolated regions of Ecuador in the 21st century, possibly driven by increased human movement, as has been documented across the Americas [60,61]. Starting near the turn of the 21^{st} century, the government of Ecuador began to heavily invest in roads and infrastructure. The new roads increased connectivity to previously isolated regions, including major cities in Esmeraldas and the Amazon, as well as to more rural areas in all provinces [62,63]. These changes may have driven the uptick in DENV FoI in Esmeraldas and the Amazon, which were not previously considered high risk areas. In Esmeraldas, connectivity first increased with Colombia, consistent with the observation that the DENV serotypes circulating in Esmeraldas between 2010-2014 more closely matched those circulating in Colombia than the rest of Ecuador [34,36,62]. A more recent study demonstrated dissemination of DENV1 and DENV2 from urban centers to more isolated communities in the Esmeraldas province [36]. Seroprevalence data from Esmeraldas demonstrates longer-term circulation of DENV and more recent introduction of CHIKV, with rural areas more likely to be only positive to DENV while those in urban areas were more likely to be exposed to both viruses, suggesting emerging pathogens take longer to reach remote areas [64]. Consistent with this observation, CHIKV caused an outbreak in rural and highly isolated communities in Esmeraldas only in 2019, long after the major chikungunya epidemic across Ecuador in 2015 [65]. As previously low-population rural areas become increasingly urbanized and connected to other rural areas and to major cities, their role in maintaining transmission and helping reseed epidemics is expected to increase [36,66]. Further, lack of infrastructure often tracks with rapid urbanization, and may result in limited water access, garbage collection, and sewage, which are all local arboviral disease risk factors in Ecuador [23-27].

In total, we found that 55.5% of Ecuador currently has favorable conditions for Aedes *aegypti*, including the provinces we identify as experiencing increases in DENV transmission intensity in the last 10 years. Notably, we also found that higher elevation regions like Pichincha/Santo Domingo and Los Ríos experienced an increase in dengue, possibly due to changes in global temperature which are predicted to expand the range of Aedes aegypti into higher elevation areas of Ecuador [30,55,67]. Our geographic models show that the coastal provinces and Amazon have high risk areas for the vector, and that population density, garbage collection, and elevation explain the most variation in vector distributions. Our estimates of high vector risk in coastal Ecuador agrees with global dengue maps and are higher than global Aedes maps; however, our estimates suggest lower risk in parts of the Andes and Amazon than predicted by some other global models [10,54,68,69] and models of Ecuador, possibly due to use of a different immature and adult mosquito dataset [30,67]. The predictors we identified as important for explaining risk partially agree with previous mapping efforts, including poor housing infrastructure and population density [54]. We also found that elevation had a strong effect. Numerous studies in Ecuador [30,55,67] and globally [10,68,69] have shown that population density, precipitation, and minimum temperature are important determinants of dengue presence and transmission intensity. In one study, the effect of elevation was small once other variables were considered [10]; given strong correlation between these variables, especially in Ecuador, it may be that they are measuring similar phenomena.

The ecological conditions in regions where DENV has recently emerged may further increase DENV intensity. For instance, while in Guayas and Manabí, entomological surveys reveal low vector density during the dry season and larvae proliferation during the wet season [33], a household survey in Esmeraldas found more complex relationships between mosquito density, rainfall, and water storage practices [70]. In the Amazon, there is not as strong a distinction between the wet and dry seasons, potentially enabling more sustained arbovirus transmission year-round. The combined effects of Amazon ecological conditions, urbanization, and population growth are evident in the Amazon cities of Iquitos, Perú and Manaus, Brazil, where once *Aedes aegypti* and DENV were introduced and the cities were large enough, the conditions were primed for severe epidemics and sustained transmission [71]. In Iquitos, Perú which is mainly accessible by river barge, invasion of the *Aedes aegypti* species was facilitated by these river boats traveling from urban to peri-urban to rural areas [72]. For this reason, similar attention should be paid to changes in the Amazon provinces of Ecuador, which are increasingly affected by population growth, urbanization, and migration due to the expansion of the oil industry [73].

Our study has several limitations. We were unable to test whether changes in Aedes risk were temporally associated with changes in DENV transmission intensity, as data on ecological and mosquito density data were not available before 2013. Further, we were unable to determine changes in the serotype-specific intensity of DENV transmission due to lack of serotype-level case data. For our FoI models, we used cases classified as Dengue Fever or Dengue Hemorrhagic Fever by the ICD code, and not those classified as other viral or hemorrhagic fevers. Given that not all cases were laboratory confirmed, it is possible that some cases caused by other diseases were accidentally counted as dengue and included in our analyses. Additionally, our analyses depended on cases detected by surveillance, which are mostly hospitalized cases. In Ecuador, as in other places, hospitalized cases underestimate disease compared to passive surveillance and depend on local healthcare provider practices [74,75]. We were also unable to quantitatively measure the effect of changes in connectivity between regions and DENV transmission intensity due to lack of available data. Due to low numbers of cases, we were unable to measure transmission changes for each separate province in the Amazon. Finally, we were unable to analyze the impact of the COVID-19 epidemic on the FoI of DENV. Data collected on dengue are likely to be highly underreported in 2020 and 2021 and may have biases in age of reporting that would alter our interpretation of the data. Dengue epidemics returned throughout Americas in 2022, as the COVID-19 pandemic waned [76]. Why this happened, and whether the COVID-19 epidemic impacted dengue has been a topic of multiple other studies and could be the topic for of a future investigation in Ecuador if appropriate data sources could be identified.

Here, we tracked the changes and drivers of DENV transmission across Ecuador using over 20 years of longitudinal case data, detailed entomological measurements, catalytic models of DENV FoI, and maximum entropy models for species distribution modeling. We show that coastal provinces with large, connected cities experiencing the highest and earliest increase in DENV transmission intensity. Areas that remained isolated like the northern coast and the Amazon regions more recently experienced a rise in DENV transmission and endemicity. However, all provinces with suitable ecological conditions were affected by the chikungunya and Zika pandemics and had regions with high *Aedes* risk. Future local and national control strategies should consider the emergence and re-emergence of multiple arboviruses transmitted by *Aedes aegypti* as well as differences in the natural and built environments, including the rapid expansion of vector-borne diseases into areas that previously were minimally affected. As human movement, population growth, urbanization, and climate change intensify, new areas within diverse countries like Ecuador will be affected [30,55,67,68]. Vector surveillance

combined with statistical prediction methods of climate and other factors can identify areas at high risk for arbovirus transmission and may serve early warning system for human dengue outbreaks [77,78]. Surveillance and vector control efforts may be further refined by learning from the historical dynamics for dengue, Zika, and chikungunya, recognizing the potential for novel emerging diseases to follow similar patterns and strategically developing control measures that consider interactions and shared risk factors for these diseases.

Acknowledgments

We thank the Ecuador Ministry of Health for providing the published dengue, Zika, and chikungunya incidence data needed for this analysis.

Author Contributions

- **Conceptualization:** Leah C. Katzelnick, Emmanuelle Quentin, Josefina Coloma, Varsovia Cevallos.
- Data curation: Emmanuelle Quentin, Patricio Ponce, Varsovia Cevallos.

Formal analysis: Leah C. Katzelnick, Emmanuelle Quentin, Varsovia Cevallos.

- Investigation: Emmanuelle Quentin, Patricio Ponce, Varsovia Cevallos.
- Methodology: Leah C. Katzelnick, Emmanuelle Quentin, Paulina Andrade, Patricio Ponce, Josefina Coloma, Varsovia Cevallos.
- Supervision: Emmanuelle Quentin, Joseph N. S. Eisenberg, Patricio Ponce, Josefina Coloma, Varsovia Cevallos.
- Writing original draft: Leah C. Katzelnick, Savannah Colston, Thien-An Ha.
- Writing review & editing: Leah C. Katzelnick, Emmanuelle Quentin, Savannah Colston, Thien-An Ha, Paulina Andrade, Joseph N. S. Eisenberg, Patricio Ponce, Josefina Coloma, Varsovia Cevallos.

References

- Bhatt S, Gething PW, Brady OJ, Messina JP, Farlow AW, Moyes CL, et al. The global distribution and burden of dengue. Nature. 2013; 496(7446):504–7. Epub 2013/04/09. https://doi.org/10.1038/ nature12060 PMID: 23563266.
- Morens DM, Fauci AS. Emerging infectious diseases: threats to human health and global stability. PLoS Pathog. 2013; 9(7):e1003467. Epub 2013/07/16. https://doi.org/10.1371/journal.ppat.1003467 PMID: 23853589.
- Brathwaite Dick O, San Martin JL, Montoya RH, del Diego J, Zambrano B, Dayan GH. The history of dengue outbreaks in the Americas. Am J Trop Med Hyg. 2012; 87(4):584–93. Epub 2012/10/09. <u>https:// doi.org/10.4269/ajtmh.2012.11-0770 PMID: 23042846</u>.
- Gubler DJ. Dengue and dengue hemorrhagic fever. Clin Microbiol Rev. 1998; 11(3):480–96. Epub 1998/07/17. https://doi.org/10.1128/CMR.11.3.480 PMID: 9665979.
- 5. World Health Organization. Global strategy for dengue prevention and control 2012–2020. World Health Organization; 2012.
- 6. DALYs GBD, Collaborators H, Murray CJ, Barber RM, Foreman KJ, Abbasoglu Ozgoren A, et al. Global, regional, and national disability-adjusted life years (DALYs) for 306 diseases and injuries and healthy life expectancy (HALE) for 188 countries, 1990–2013: quantifying the epidemiological transition. Lancet. 2015; 386(10009):2145–91. Epub 2015/09/01. https://doi.org/10.1016/S0140-6736(15)61340-X PMID: 26321261.
- Yactayo S, Staples JE, Millot V, Cibrelus L, Ramon-Pardo P. Epidemiology of Chikungunya in the Americas. The Journal of Infectious Diseases. 2016; 214(suppl_5):S441–S5. https://doi.org/10.1093/infdis/ jiw390 PMID: 27920170

- Zhang Q, Sun K, Chinazzi M, Pastore y Piontti A, Dean NE, Rojas DP, et al. Spread of Zika virus in the Americas. Proceedings of the National Academy of Sciences. 2017; 114(22):E4334–E43. <u>https://doi.org/10.1073/pnas.1620161114</u> PMID: 28442561
- Rodriguez-Barraquer I, Salje H, Cummings DA. Opportunities for improved surveillance and control of dengue from age-specific case data. Elife. 2019; 8. Epub 2019/05/24. <u>https://doi.org/10.7554/eLife.</u> 45474 PMID: 31120419.
- Cattarino L, Rodriguez-Barraquer I, Imai N, Cummings DAT, Ferguson NM. Mapping global variation in dengue transmission intensity. Sci Transl Med. 2020; 12(528). Epub 2020/01/31. <u>https://doi.org/10.1126/scitranslmed.aax4144 PMID: 31996463</u>.
- Cummings DA, Iamsirithaworn S, Lessler JT, McDermott A, Prasanthong R, Nisalak A, et al. The impact of the demographic transition on dengue in Thailand: insights from a statistical analysis and mathematical modeling. PLoS Med. 2009; 6(9):e1000139. Epub 2009/09/02. <u>https://doi.org/10.1371/journal.</u> pmed.1000139 PMID: 19721696.
- Imai N, Dorigatti I, Cauchemez S, Ferguson NM. Estimating Dengue Transmission Intensity from Case-Notification Data from Multiple Countries. PLoS Negl Trop Dis. 2016; 10(7):e0004833. Epub 20160711. https://doi.org/10.1371/journal.pntd.0004833 PMID: 27399793.
- Rodriguez-Barraquer I, Cordeiro MT, Braga C, de Souza WV, Marques ET, Cummings DA. From reemergence to hyperendemicity: the natural history of the dengue epidemic in Brazil. PLoS Negl Trop Dis. 2011; 5(1):e935. Epub 2011/01/20. https://doi.org/10.1371/journal.pntd.0000935 PMID: 21245922.
- Fischer DB, Halstead SB. Observations related to pathogenesis of dengue hemorrhagic fever. V. Examination of agspecific sequential infection rates using a mathematical model. Yale J Biol Med. 1970; 42(5):329–49. PMID: 5419207.
- Ferguson NM, Donnelly CA, Anderson RM. Transmission dynamics and epidemiology of dengue: insights from age-stratified sero-prevalence surveys. Philos Trans R Soc Lond B Biol Sci. 1999; 354 (1384):757–68. Epub 1999/06/12. https://doi.org/10.1098/rstb.1999.0428 PMID: 10365401.
- Schenzle D, Dietz K, Frosner GG. Antibody against hepatitis A in seven European countries. II. Statistical analysis of cross-sectional surveys. Am J Epidemiol. 1979; 110(1):70–6. Epub 1979/07/01. https://doi.org/10.1093/oxfordjournals.aje.a112790 PMID: 313700.
- Katzelnick LC, Ben-Shachar R, Mercado JC, Rodriguez-Barraquer I, Elizondo D, Arguello S, et al. Dynamics and determinants of the force of infection of dengue virus from 1994 to 2015 in Managua, Nicaragua. Proc Natl Acad Sci U S A. 2018; 115(42):10762–7. Epub 2018/09/30. <u>https://doi.org/10.1073/</u> pnas.1809253115 PMID: 30266790.
- Huang AT, Takahashi S, Salje H, Wang L, Garcia-Carreras B, Anderson K, et al. Assessing the role of multiple mechanisms increasing the age of dengue cases in Thailand. Proc Natl Acad Sci U S A. 2022; 119(20):e2115790119. Epub 20220509. https://doi.org/10.1073/pnas.2115790119 PMID: 35533273.
- Reiner RC Jr., Stoddard ST, Forshey BM, King AA, Ellis AM, Lloyd AL, et al. Time-varying, serotypespecific force of infection of dengue virus. Proc Natl Acad Sci U S A. 2014; 111(26):E2694–702. Epub 20140520. https://doi.org/10.1073/pnas.1314933111 PMID: 24847073.
- Rodriguez-Barraquer I, Buathong R, Iamsirithaworn S, Nisalak A, Lessler J, Jarman RG, et al. Revisiting Rayong: shifting seroprofiles of dengue in Thailand and their implications for transmission and control. Am J Epidemiol. 2014; 179(3):353–60. Epub 20131105. https://doi.org/10.1093/aje/kwt256 PMID: 24197388.
- Egger JR, Ooi EE, Kelly DW, Woolhouse ME, Davies CR, Coleman PG. Reconstructing historical changes in the force of infection of dengue fever in Singapore: implications for surveillance and control. Bull World Health Organ. 2008; 86(3):187–96. https://doi.org/10.2471/blt.07.040170 PMID: 18368205.
- Marcondes CB, Contigiani M, Gleiser RM. Emergent and Reemergent Arboviruses in South America and the Caribbean: Why So Many and Why Now? J Med Entomol. 2017; 54(3):509–32. <u>https://doi.org/ 10.1093/jme/tjw209</u> PMID: 28399216.
- Stewart Ibarra AM, Ryan SJ, Beltran E, Mejia R, Silva M, Munoz A. Dengue vector dynamics (Aedes aegypti) influenced by climate and social factors in Ecuador: implications for targeted control. PLoS One. 2013; 8(11):e78263. Epub 2013/12/11. https://doi.org/10.1371/journal.pone.0078263 PMID: 24324542.
- Stewart-Ibarra AM, Munoz AG, Ryan SJ, Ayala EB, Borbor-Cordova MJ, Finkelstein JL, et al. Spatiotemporal clustering, climate periodicity, and social-ecological risk factors for dengue during an outbreak in Machala, Ecuador, in 2010. BMC Infect Dis. 2014; 14:610. Epub 20141125. <u>https://doi.org/10.1186/</u> s12879-014-0610-4 PMID: 25420543.
- Ha TA, Leon TM, Lalangui K, Ponce P, Marshall JM, Cevallos V. Household-level risk factors for Aedes aegypti pupal density in Guayaquil, Ecuador. Parasit Vectors. 2021; 14(1):458. Epub 20210907. <u>https:// doi.org/10.1186/s13071-021-04913-0 PMID: 34493321.</u>

- 26. Lippi CA, Stewart-Ibarra AM, Endy TP, Abbott M, Cueva C, Heras F, et al. Exploring the utility of socialecological and entomological risk factors for dengue infection as surveillance indicators in the dengue hyper-endemic city of Machala, Ecuador. PLoS Negl Trop Dis. 2021; 15(3):e0009257. Epub 20210319. https://doi.org/10.1371/journal.pntd.0009257 PMID: 33740003.
- Martin JL, Lippi CA, Stewart-Ibarra AM, Ayala EB, Mordecai EA, Sippy R, et al. Household and climate factors influence Aedes aegypti presence in the arid city of Huaquillas, Ecuador. PLoS Negl Trop Dis. 2021; 15(11):e0009931. Epub 20211116. https://doi.org/10.1371/journal.pntd.0009931 PMID: 34784348.
- Lippi CA, Stewart-Ibarra AM, Munoz AG, Borbor-Cordova MJ, Mejia R, Rivero K, et al. The Social and Spatial Ecology of Dengue Presence and Burden during an Outbreak in Guayaquil, Ecuador, 2012. Int J Environ Res Public Health. 2018; 15(4). Epub 20180423. <u>https://doi.org/10.3390/ijerph15040827</u> PMID: 29690593.
- Castillo KC, Körbl B, Stewart A, Gonzalez JF, Ponce F. Application of spatial analysis to the examination of dengue fever in Guayaquil, Ecuador. Procedia Environmental Sciences. 2011; 7:188–93. <u>https:// doi.org/10.1016/j.proenv.2011.07.033</u>
- Lippi CA, Stewart-Ibarra AM, Loor M, Zambrano JED, Lopez NAE, Blackburn JK, et al. Geographic shifts in Aedes aegypti habitat suitability in Ecuador using larval surveillance data and ecological niche modeling: Implications of climate change for public health vector control. PLoS Negl Trop Dis. 2019; 13 (4):e0007322. Epub 20190417. https://doi.org/10.1371/journal.pntd.0007322 PMID: 30995228.
- Connor ME. Yellow fever control in Ecuador: Preliminary report. Journal of the American Medical Association. 1920; 74(10):650–1. https://doi.org/10.1001/jama.1920.02620100010004
- Izurieta RO, Macaluso M, Watts DM, Tesh RB, Guerra B, Cruz LM, et al. Assessing yellow Fever risk in the ecuadorian Amazon. J Glob Infect Dis. 2009; 1(1):7–13. https://doi.org/10.4103/0974-777X.49188 PMID: 20300380.
- Real-Cotto JJ, Regato Arrata ME, Burgos Yépez VE, Jurado Cobeña ET. Evolución del virus dengue en el Ecuador: Período 2000 a 2015. Anales de la Facultad de Medicina. 2017; 78:29–35.
- Márquez S, Carrera J, Espín E, Cifuentes SG, Trueba G, Coloma J, et al. Dengue Serotype Differences in Urban and Semi-Rural Communities in Ecuador. Avances en Ciencias e Ingeniería. 2018; 10. https:// doi.org/10.18272/aci.v10i1.959
- Maljkovic Berry I, Rutvisuttinunt W, Sippy R, Beltran-Ayala E, Figueroa K, Ryan S, et al. The origins of dengue and chikungunya viruses in Ecuador following increased migration from Venezuela and Colombia. BMC Evolutionary Biology. 2020; 20(1):31. https://doi.org/10.1186/s12862-020-1596-8 PMID: 32075576
- Márquez S, Lee G, Gutiérrez B, Bennett S, Coloma J, Eisenberg JNS, et al. Phylogenetic Analysis of Transmission Dynamics of Dengue in Large and Small Population Centers, Northern Ecuador. Emerging Infectious Disease journal. 2023; 29(5):888. https://doi.org/10.3201/eid2905.221226 PMID: 37080979
- Stewart-Ibarra AM, Ryan SJ, Kenneson A, King CA, Abbott M, Barbachano-Guerrero A, et al. The Burden of Dengue Fever and Chikungunya in Southern Coastal Ecuador: Epidemiology, Clinical Presentation, and Phylogenetics from the First Two Years of a Prospective Study. Am J Trop Med Hyg. 2018; 98 (5):1444–59. Epub 20180301. https://doi.org/10.4269/ajtmh.17-0762 PMID: 29512482.
- Instituto Nacional de Estadística y Censos [Internet]. <u>https://www.ecuadorencifras.gob.ec/camas-y-egresos-hospitalarios/.</u>
- Dirección Nacional de Vigilancia Epidemiológica: Anuario Epidemiológico [Internet]. Ministerio de Salud Pública de Ecuador. https://www.salud.gob.ec/direccion-nacional-de-vigilancia-epidemiologica/.
- 40. Sistema Nacional de Vigilancia en Salud Pública Ecuador [Internet]. https://viepi.msp.gob.ec/login.php.
- 41. Clasificador geográfico estadístico–Archivos shapefile de la División Político Administrativa a nivel nacional del año 2012 [Internet]. Instituto Nacional de Estadística y Censos–INEC. https://www. ecuadorencifras.gob.ec/documentos/web-inec/Geografia_Estadistica/Micrositio_geoportal/index. html#clasificador-geog-dpa.
- 42. Cleveland WS, Grosse E, Shyu WM. Local regression models. In: Chambers JM, Hastie TJ, editors. Statistical Models in S: Wadsworth & Brooks/Cole; 1992.
- Jacoby WG. Loess: a nonparametric, graphical tool for depicting relationships between variables. Electoral Studies. 2000; 19(4):577–613. https://doi.org/10.1016/S0261-3794(99)00028-1
- Sangkawibha N, Rojanasuphot S, Ahandrik S, Viriyapongse S, Jatanasen S, Salitul V, et al. Risk factors in dengue shock syndrome: a prospective epidemiologic study in Rayong, Thailand. I. The 1980 outbreak. Am J Epidemiol. 1984; 120(5):653–69. <u>https://doi.org/10.1093/oxfordjournals.aje.a113932</u> PMID: 6496446.
- Halstead SB, Scanlon JE, Umpaivit P, Udomsakdi S. Dengue and chikungunya virus infection in man in Thailand, 1962–1964. IV. Epidemiologic studies in the Bangkok metropolitan area. Am J Trop Med Hyg. 1969; 18(6):997–1021. https://doi.org/10.4269/ajtmh.1969.18.997 PMID: 4390977.

- Kouri GP, Guzman MG, Bravo JR, Triana C. Dengue haemorrhagic fever/dengue shock syndrome: lessons from the Cuban epidemic, 1981. Bull World Health Organ. 1989; 67(4):375–80. PMID: 2805215.
- Olkowski S, Forshey BM, Morrison AC, Rocha C, Vilcarromero S, Halsey ES, et al. Reduced risk of disease during postsecondary dengue virus infections. J Infect Dis. 2013; 208(6):1026–33. Epub 2013/06/ 19. https://doi.org/10.1093/infdis/jit273 PMID: 23776195.
- **48.** Stan Development Team. RStan: the R interface to Stan. 2020.
- R Core Team. R: A language and environment for statistical computing. R Foundation for Statistical Computing. Vienna, Austria: 2022.
- 50. MaxEnt (version 3.3.3). https://biodiversityinformatics.amnh.org/open_source/maxent/.
- Phillips SJ, Anderson RP, Schapire RE. Maximum entropy modeling of species geographic distributions. Ecol Model. 2006; 190(3–4):231–59. https://doi.org/10.1016/j.ecolmodel.2005.03.026
- Phillips SJ, Dudik M. Modeling of species distributions with Maxent: new extensions and a comprehensive evaluation. Ecography. 2008; 31(2):161–75. <u>https://doi.org/10.1111/j.0906-7590.2008</u>.
 5203.x
- Carrazco-Montalvo A, Ponce P, Villota SD, Quentin E, Muñoz-Tobar S, Coloma J, et al. Establishment, Genetic Diversity, and Habitat Suitability of Aedes albopictus Populations from Ecuador. Insects. 2022; 13(3):305. https://doi.org/10.3390/insects13030305 PMID: 35323603
- Louis VR, Phalkey R, Horstick O, Ratanawong P, Wilder-Smith A, Tozan Y, et al. Modeling tools for dengue risk mapping—a systematic review. Int J Health Geogr. 2014; 13:50. Epub 20141209. <u>https:// doi.org/10.1186/1476-072X-13-50 PMID: 25487167</u>.
- 55. Jacome G, Vilelaa P, Yoo C. Social-ecological modelling of the spatial distribution of dengue fever and its temporal dynamics in Guayaquil, Ecuador for climate change adaption. Ecol Inform. 2019; 49:1–12. https://doi.org/10.1016/j.ecoinf.2018.11.001
- 56. Yin S, Ren C, Shi Y, Hua J, Yuan HY, Tian LW. A Systematic Review on Modeling Methods and Influential Factors for Mapping Dengue-Related Risk in Urban Settings. Int J Environ Res Public Health. 2022; 19(22). Epub 20221118. https://doi.org/10.3390/ijerph192215265 PMID: 36429980.
- 57. EarthData [Internet]. NASA. https://www.earthdata.nasa.gov/.
- Anderson RM, May RM. Infectious Diseases of Humans Dynamics and Control. Oxford: Oxford University Press; 1991.
- Messina JP, Brady OJ, Scott TW, Zou C, Pigott DM, Duda KA, et al. Global spread of dengue virus types: mapping the 70 year history. Trends Microbiol. 2014; 22(3):138–46. Epub 2014/01/29. https://doi.org/10.1016/j.tim.2013.12.011 PMID: 24468533.
- Allicock OM, Sahadeo N, Lemey P, Auguste AJ, Suchard MA, Rambaut A, et al. Determinants of dengue virus dispersal in the Americas. Virus Evol. 2020; 6(2):veaa074. Epub 20201202. <u>https://doi.org/10. 1093/ve/veaa074</u> PMID: 33408877.
- Gutierrez B, da Silva Candido D, Bajaj S, Rodriguez Maldonado AP, Ayala FG, Rodriguez MLT, et al. Convergent trends and spatiotemporal patterns of Aedes-borne arboviruses in Mexico and Central America. PLoS Negl Trop Dis. 2023; 17(9):e0011169. Epub 20230906. https://doi.org/10.1371/journal. pntd.0011169 PMID: 37672514.
- 62. Eisenberg JNS, Cevallos W, Ponce K, Levy K, Bates SJ, Scott JC, et al. Environmental change and infectious disease: How new roads affect the transmission of diarrheal pathogens in rural Ecuador. Proceedings of the National Academy of Sciences. 2006; 103(51):19460–5. https://doi.org/10.1073/pnas. 0609431104 PMID: 17158216
- Zelner JL, Trostle J, Goldstick JE, Cevallos W, House JS, Eisenberg JN. Social connectedness and disease transmission: social organization, cohesion, village context, and infection risk in rural Ecuador. Am J Public Health. 2012; 102(12):2233–9. Epub 20121018. <u>https://doi.org/10.2105/AJPH.2012.300795</u> PMID: 23078481.
- Chis Ster I, Rodriguez A, Romero NC, Lopez A, Chico M, Montgomery J, et al. Age-dependent seroprevalence of dengue and chikungunya: inference from a cross-sectional analysis in Esmeraldas Province in coastal Ecuador. BMJ Open. 2020; 10(10):e040735. Epub 2020/10/18. https://doi.org/10.1136/ bmjopen-2020-040735 PMID: 33067302.
- Marquez S, Lee GO, Andrade P, Zuniga J, Trueba G, Eisenberg JNS, et al. A Chikungunya Outbreak in a Dengue-endemic Region in Rural Northern Coastal Ecuador. Am J Trop Med Hyg. 2022; 107 (6):1226–33. Epub 20221114. https://doi.org/10.4269/ajtmh.22-0296 PMID: 36375454.
- Lana RM, Gomes MFdC, Lima TFMd, Honório NA, Codeço CT. The introduction of dengue follows transportation infrastructure changes in the state of Acre, Brazil: A network-based analysis. PLOS Neglected Tropical Diseases. 2017; 11(11):e0006070. https://doi.org/10.1371/journal.pntd.0006070 PMID: 29149175

- Escobar LE, Romero-Alvarez D, Leon R, Lepe-Lopez MA, Craft ME, Borbor-Cordova MJ, et al. Declining Prevalence of Disease Vectors Under Climate Change. Sci Rep. 2016; 6:39150. Epub 20161216. https://doi.org/10.1038/srep39150 PMID: 27982119.
- Messina JP, Brady OJ, Golding N, Kraemer MUG, Wint GRW, Ray SE, et al. The current and future global distribution and population at risk of dengue. Nat Microbiol. 2019; 4(9):1508–15. Epub 2019/06/ 12. https://doi.org/10.1038/s41564-019-0476-8 PMID: 31182801.
- Kraemer MU, Sinka ME, Duda KA, Mylne AQ, Shearer FM, Barker CM, et al. The global distribution of the arbovirus vectors Aedes aegypti and Ae. albopictus. Elife. 2015; 4:e08347. Epub 20150630. <u>https:// doi.org/10.7554/eLife.08347 PMID: 26126267.</u>
- 70. Lee GO, Vasco L, Marquez S, Zuniga-Moya JC, Van Engen A, Uruchima J, et al. A dengue outbreak in a rural community in Northern Coastal Ecuador: An analysis using unmanned aerial vehicle mapping. PLoS Negl Trop Dis. 2021; 15(9):e0009679. Epub 20210927. <u>https://doi.org/10.1371/journal.pntd.</u> 0009679 PMID: 34570788.
- de Castro DB, Sampaio VS, de Albuquerque BC, Pinto RC, Sadahiro M, Dos Passos RA, et al. Dengue epidemic typology and risk factors for extensive epidemic in Amazonas state, Brazil, 2010–2011. BMC Public Health. 2018; 18(1):356. Epub 20180315. <u>https://doi.org/10.1186/s12889-018-5251-x</u> PMID: 29544456.
- 72. Guagliardo SA, Morrison AC, Barboza JL, Requena E, Astete H, Vazquez-Prokopec G, et al. River Boats Contribute to the Regional Spread of the Dengue Vector Aedes aegypti in the Peruvian Amazon. PLOS Neglected Tropical Diseases. 2015; 9(4):e0003648. <u>https://doi.org/10.1371/journal.pntd.</u> 0003648 PMID: 25860352
- **73.** Rights Violations in the Ecuadorian Amazon: The Human Consequences of Oil Development. Health Hum Rights. 1994; 1(1):82–100. PMID: <u>10395712</u>.
- 74. Handel AS, Ayala EB, Borbor-Cordova MJ, Fessler AG, Finkelstein JL, Espinoza RXR, et al. Knowledge, attitudes, and practices regarding dengue infection among public sector healthcare providers in Machala, Ecuador. Trop Dis Travel Med Vaccines. 2016; 2:8. Epub 20160601. <u>https://doi.org/10.1186/</u> s40794-016-0024-y PMID: 28883952.
- Vitale M, Lupone CD, Kenneson-Adams A, Ochoa RJ, Ordonez T, Beltran-Ayala E, et al. A comparison of passive surveillance and active cluster-based surveillance for dengue fever in southern coastal Ecuador. BMC Public Health. 2020; 20(1):1065. Epub 20200706. https://doi.org/10.1186/s12889-020-09168-5 PMID: 32631315.
- 76. Disease Outbreak News: Dengue–the Region of the Americas: World Health Organization. https:// www.who.int/emergencies/disease-outbreak-news/item/2023-DON475#:~:text=Compared%20to% 20the%20last%20epidemiological,126%20431%20cases%20year%202023
- Cevallos V, Ponce P, Waggoner JJ, Pinsky BA, Coloma J, Quiroga C, et al. Zika and Chikungunya virus detection in naturally infected Aedes aegypti in Ecuador. Acta Tropica. 2018; 177:74–80. https://doi.org/ 10.1016/j.actatropica.2017.09.029 PMID: 28982578
- Stewart-Ibarra AM, Lowe R. Climate and non-climate drivers of dengue epidemics in southern coastal ecuador. Am J Trop Med Hyg. 2013; 88(5):971–81. Epub 2013/03/13. https://doi.org/10.4269/ajtmh.12-0478 PMID: 23478584.